Author’s response to reviews

Title: Association Between Basilar Artery Configuration and Vessel Wall Features: A Prospective High-resolution Magnetic Resonance Imaging Study

Authors:
Ziqi Xu (zyxuziqi@zju.edu.cn)
Mingyao Li (1020989554@qq.com)
Zhikai Hou (houzhikai92@163.com)
Jinhao Lyu (330322990@qq.com)
Na Zhang (278382341@qq.com)
Xin Lou (xinlou301@sina.com)
Zhongrong Miao (zhongrongm@163.com)
Ning Ma (maning_03@hotmail.com)

Version: 2 Date: 10 Oct 2019

Author’s response to reviews:

Dear editors,

We confirm that the revised manuscripts conforms to the journal style and do not change the author.

We have added the statement of the availability of the data and section. The datasets used and analysed during the current study are available from the corresponding author on reasonable request.

We ensure that all figures, tables and additional/supplementary files are cited within the text.

Before submission, the language of our manuscript was edited by English language editing service.

sincerely,

Ziqi Xu